Author’s response to reviews

Title: Economic burden of symptomatic iron deficiency - A Survey among Swiss women

Authors:

Patricia Blank (patricia.blank@unibas.ch)
Yuki Tomonaga (yuki.tomonaga@uzh.ch)
Thomas Szucs (thomas.szucs@unibas.ch)
Matthias Schwenkglenks (matthias.schwenkglenks@uzh.ch)

Version: 2 Date: 07 Aug 2017

Author’s response to reviews:

Dear Editor

Please find attached our revised version of the manuscript entitled "Economic burden of symptomatic iron deficiency – A Survey among Swiss women ". The revision includes an amended Statement of the "declarations" regarding the availability of raw data. We hope that this Statement fulfils your editorial policies and the manuscript can now be sent out for review.

Sincerely yours

Patricia Blank